

# Significant amplification of instantaneous extreme precipitation with convective self-aggregation

Nicolas A. da Silva, Caroline Muller, Sara Shamekh, Benjamin Fildier

# ▶ To cite this version:

Nicolas A. da Silva, Caroline Muller, Sara Shamekh, Benjamin Fildier. Significant amplification of instantaneous extreme precipitation with convective self-aggregation. Journal of Advances in Modeling Earth Systems, 2021, 10.1029/2021ms002607. hal-03385132

# HAL Id: hal-03385132 https://hal.sorbonne-universite.fr/hal-03385132v1

Submitted on 19 Oct 2021

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

# Significant amplification of instantaneous extreme precipitation with convective self-aggregation

Nicolas A. Da Silva<sup>1</sup>, Caroline Muller<sup>2</sup>, Sara Shamekh<sup>3</sup>, Benjamin Fildier<sup>2</sup>

 $^1\mathrm{Complexity}$  and Climate, Leibniz Centre for Tropical Marine Research, Fahrenheitstrasse 6, 28359,

Bremen, Germany <sup>2</sup>Laboratoire de Météorologie Dynamique (LMD)/Institut Pierre Simon Laplace (IPSL), École Normale Supérieure, Paris Sciences & Lettres (PSL) Research University, Sorbonne Université, École Polytechnique, CNRS, F-75005 Paris, France <sup>3</sup>Columbia University, New York 10032, USA

#### **Key Points:**

- Convective self-aggregation may increase both accumulated and instantaneous rainfall extremes
- This increased precipitation is related to reduced rain re-evaporation and enhanced accretion efficiency with convective self-aggregation
- Extreme convective updrafts within self-aggregated convection are weaker due to warmer environment

Corresponding author: Nicolas Da Silva, dasilvanicolas950gmail.com

This article has been accepted for publication  $and^{L}$  undergone full peer review but has not been through the copyediting, typesetting, pagination and proofreading process, which may lead to differences between this version and the Version of Record. Please cite this article as doi: 10.1029/2021MS002607.

#### Abstract

This work explores the effect of convective self-aggregation on extreme rainfall intensities through an analysis at several stages of the cloud lifecycle. In addition to increases in 3-hourly extremes consistent with previous studies, we find that instantaneous rainrates increase significantly (+30%). We mainly focus on instantaneous extremes and, using a recent framework, relate their increase to increased precipitation efficiency: the local increase in relative humidity drives larger accretion efficiency and lower re-evaporation. An in-depth analysis based on an adapted scaling for precipitation extremes reveals that the dynamic contribution decreases (-25%) while the thermodynamic is slightly enhanced (+5%) with convective self-aggregation, leading to lower condensation rates. When the atmosphere is more organized into a moist convecting region and a dry convection-free region, deep convective updrafts are surrounded by a warmer environment which reduces convective instability and thus the dynamic contribution. The moister boundary-layer explains the positive thermodynamic contribution. The microphysic contribution is increased by +50 % with aggregation. The latter is partly due to reduced evaporation of rain falling through a moister near-cloud environment, but also to the associated larger accretion efficiency. Thus, a potential change in convective organization regimes in a warming climate could lead to an evolution of tropical precipitation extremes significantly different than that expected from thermodynamical considerations. The relevance of self-aggregation to the real tropics is still debated. Improved fundamental understanding of self-aggregation, its sensitivity to warming and connection to precipitation extremes, is hence crucial to achieve accurate rainfall projections in a warming climate.

## Plain Language Summary

Heavy precipitation and floods are frequent in the tropics. The spatial organization of weather systems is often associated with these events. Our study investigates the case of convective self-aggregation which is a particular type of cloud systems' organization observed in idealized numerical simulations. We find that convective self-aggregation tends to increase rainfall intensities by 30% to 70%. There are several processes involved in the formation of heavy rainfall: vertical motion air, condensation into cloud droplets, growth of these droplets into precipitating drops, collection of other cloud-droplets through their descent and partial re-evaporation between cloud base and the ground. We examine the contribution of each of these processes and find that the increase in rain rates with convective self-aggregation is related to both lower rain re-evaporation and more efficient cloud droplet collection by rain drops through their descent. It is still unclear how heavy rainfall will evolve in a warming climate. While the relationship between temperature and water vapor suggests that heavy rainfall will increase by 7% per 1K, our result shows that a hypothetic change in the organization of weather systems could potentially lead to more dramatic changes in heavy rainfall in a future warming climate.

## 1 Introduction

A large uncertainty regarding tropical precipitation and its evolution in a warming climate is related to convective organization and its response to warming. The spatial organization of deep convection at mesoscales, i.e. hundreds of kilometers, is ubiquitous in the tropics in the form of Mesoscale Convective Systems embedded in larger scale perturbations such as the Madden-Julian Oscillation (Jiang et al., 2020). Convective organization has important societal impacts, notably as it is associated with extreme precipitation (Rossow et al., 2013). A recent observational study (Tan et al., 2015) finds that recent trends in tropical precipitation can be linked to changes in the frequency of occurrence of organized mesoscale cloud systems. Here we will briefly mention the different modes of organization and review the mechanisms involved in the amplification of heavy rainfall with warming – with and without organization –, before focusing on an highly idealized case of organization in this study, convective self-aggregation.

Mesoscale convective organization can be forced by dynamical processes on large scales (namely, scales similar or larger than mesoscales), for instance vertical wind shear

or land-ocean contrasts and can take the form of mesoscale convective systems or squall lines. It can also arise from internal feedbacks linked to the interaction of clouds with their environment, in the form of large and moist convective areas surrounded by dry subsiding areas (Bretherton et al., 2005; Muller & Held, 2012). In the latter case, spontaneous inhomogeneous spatial clustering of clouds is triggered in otherwise homogeneous unforced environments, a process called self-aggregation of convection, mainly diagnosed in models so far (Wing et al., 2017). It is important to note that this "clean" mode of organization (in large environments deprived from wind shear, rotation and sea-surface temperature (SST) gradients) can be far from most situations currently observable, but such a self-aggregated state is argued to be an alternate stable state of radiative-convective equilibrium (RCE) that could be reached by the tropics as a whole in a warmer future (Emanuel et al., 2014). Links were also made between some of the processes involved in self-aggregation simulations and those observed in the real world (Holloway et al., 2017). A natural question that we address here is then: how could precipitation extremes be impacted by selfaggregation, and why? We will address it using idealized cloud-resolving simulations run in non-rotating RCE, and in the absence of wind shear. This "clean" modeling setup will allow a robust attribution of the processes involved, in order to serve as a starting point for integrating more complexity and realism in future modeling experiments.

Several studies have investigated the response of precipitation extremes to warming using idealized simulations of disorganized convection. Two different cloud-resolving models (CRMs) showed increases in precipitation extremes close to low-tropospheric moisture (Muller et al., 2011; Romps, 2011). The fact that precipitation extremes follow low-tropospheric humidity and not column-integrated humidity can be understood using a theoretical scaling for precipitation extremes, first introduced in Betts (1987) and O'Gorman and Schneider (2009a), and refined to connect it to microphysics (Muller et al., 2011; Muller & Takayabu, 2020). It relates the changes of precipitation extremes to three contributions: a thermodynamic contribution related to water vapor, a dynamic contribution related to vertical mass flux in extreme updrafts, and a microphysic contribution related to precipitation efficiency. The thermodynamic component, dominant in disorganized convection, is not always the prevailing term when convection is organized.

Mechanisms advanced for changes in extremes are sensitive to the mode of organization analyzed. In idealized squall line simulations, Singleton and Toumi (2013) find an amplification of precipitation extremes exceeding significantly the thermodynamic theoretical expectation, when the warming is uniform in height. In the tropics, one expects the atmosphere to warm following a moist adiabat, with larger warming aloft. If instead the warming is uniform with height, a situation which might be more relevant to the mid latitudes, atmospheric instability is enhanced (Loriaux et al., 2013). Consistently, faster updrafts contribute positively to the dynamic contribution, yielding a larger amplification of precipitation extremes with warming than the thermodynamic contribution would entail (Attema et al., 2014; Singleton & Toumi, 2013). We note though that the link between stability and precipitation extremes is not necessarily straightforward. Differences in atmospheric stability may result in differences in convection, but extremes in precipitation intensity do not necessarily follow the atmospheric stability (Hamada et al., 2015).

The increase of precipitation extremes with warming for a given degree of convective organization was also investigated in Muller (2013) but with a warming following the moist adiabat, as expected in tropical atmospheres. In that study, vertical wind shear of varying amplitude is used to organize convection into squall lines. As well as for disorganized convection, precipitation extremes also increase at about 7% K<sup>-1</sup> of warming for a given degree of convective organization, consistent with the thermodynamic theoretical expectation from the Clausius-Clapeyron equation (Held & Soden, 2006). However, a change in the degree of convective organization can lead to up to a doubling of extreme rainfall rates. Thus the increase of precipitation extremes from a change in convective organization is larger than that associated with warming by several degrees.

In the case of self-aggregation, it appears useful to separate the role of warming from the occurrence of aggregation itself when studying the response of extreme rain. Pendergrass et al. (2016) document an SST threshold over which convection selfaggregates, and the change from unorganized to self-aggregated convection is associated with an increase of precipitation extremes largely exceeding the Clausius Clapeyron law (super-CC). Similarly, using CRM experiments, Fildier et al. (2020) also observe a jump in precipitation extremes from disorganized to aggregated simulations. Moreover, they note that super-CC regimes can occur within aggregated convection but not within the disorganized convection. They link these super-CC rates to changes in precipitation efficiency. Bao and Sherwood (2019, BS19) also investigate self-aggregation and precipitation extremes, using a different cloud-resolving model (WRF). They find that self-aggregation has a small impact on extreme instantaneous precipitation, but affects strongly daily precipitation extremes. Similarly to Fildier et al. (2020), they note that extreme instantaneous precipitation is more sensitive to microphysical processes. Our work builds on these recent studies, using the theoretical scaling (Muller & Takayabu, 2020) to further investigate and quantify the physical processes associated to the change of precipitation extremes with self-aggregation. Of particular interest are the following questions:

- Does convective self-aggregation only affect rainfall intensities accumulated over time, or also instantaneous rain rates?
- Which contribution mainly explains the effect of aggregation on extreme rain rates? Is it the thermodynamic, dynamic, or microphysic contribution?
- Through which physical mechanisms can the change in precipitation efficiency affect rain extremes under convective self-aggregation?

The next section (§ 2) describes the cloud-resolving model and simulations used. Results are presented in § 3. The thermodynamic and dynamic contributions to extreme precipitation are analyzed in § 4, followed by the microphysic contribution in § 5. Conclusions, as well as key implications of our results in a warming climate and outstanding open questions, are discussed in §6.

## 2 Cloud-resolving simulations

#### 2.1 Cloud-resolving model

The cloud-resolving model used here is the System for Atmospheric Modeling, or SAM. This model solves the anelastic conservation equations for momentum, total water and energy. There are six water species in SAM (vapor, liquid cloud water, cloud ice, liquid rain, graupel, snow). The prognostic thermodynamic variables of the model are liquid/ice water static energy, total non-precipitating water (vapor, liquid cloud water, cloud water, cloud ice), and total precipitating water (liquid rain, graupel, snow). The longwave and shortwave radiative cooling rates are computed using the radiation code from the National Center for Atmospheric Research (NCAR) Community Atmosphere Model version 3 (CAM3; Collins et al. (2006)). As in numerous previous studies of self-aggregation with SAM, we use the 1-moment microphysics package. Further details about the model can be found in Khairoutdinov and Randall (2003).

All the runs are in doubly-periodic geometry starting from homogeneous initial conditions, and with an imposed sea-surface temperature of 300 K. The resolution is 3 km and the domain size 768 km in both horizontal directions. This relatively small domain size and low resolution do not allow to resolve all convective processes and dynamics and their nonlinear interactions over a wide spectrum range, only from tens to hundreds of kilometers. The vertical grid has 64 levels with the first level at 37.5 m and grid spacing gradually increasing from 80 m near the surface to 400 m above 5 km. To reduce gravity wave reflection and buildup, Newtonian damping is applied to all prognostic variables in the upper third of the model domain (18 to 27 km altitude).

#### 2.2 Simulations

We perform two simulations. The first (CTRL) with interactive radiative cooling rates (calculated everywhere in the domain at each time step), but these are horizontally homogenized at each time step and height in order to remove radiative feedbacks and thus prevent convective self-aggregation. In this simulation, convection is somewhat randomly distributed, and precipitating events can occur everywhere in the domain, with convection resembling "pop-corn" convection (Fig. 1a,b). Such simulations are occasionally referred to as disorganized RCE. **Figure 1.** Occurrence frequency of rain over the domain for CTRL (a, b) and AGG (c, d) simulations and for both instantaneous (a, c) and 3-hourly (b, d) precipitations.

The second simulation (AGG) has fully-interactive radiation, namely radiative cooling rates are resolved in 3D space and applied to the temperature tendency at each time step, and self-aggregates. Conditions favoring convective organization are still a subject of active research, but large domains and the presence of a radiative feedback on the atmospheric circulation are often identified as sufficient conditions for self-aggregation in models (Muller & Held, 2012; Muller & Bony, 2015). In that case, convection and precipitating events are confined to the moist circular region of the domain (Fig. 1c,d).

In both cases, the model is run to non-rotating radiative-convective equilibrium, or RCE, reached after about 50 days, and the simulations are run for 100 days. Only the last 30 days of the simulations are used for our analysis (day 70 until day 100). Non-rotating RCE is an idealization of the tropical atmosphere, in which the Earth's rotation is neglected (a reasonable approximation in the tropics where the Coriolis parameter is small), and in which the large-scale motion (larger than the domain) is neglected. Thus there is no import or export of moist-static energy into or out of the domain, and in the domain-mean, the net atmospheric radiative cooling (top of atmosphere minus surface) balances the input of energy into the atmosphere at the surface, namely latent and sensible heat fluxes. The solar irradiance is fixed to 413 W.m<sup>-2</sup> (same as Shamekh et al. (2020)), therefore removing diurnal oscillations of convection in the simulations.

# 3 Impact of convective aggregation on precipitation extremes

#### 3.1 Significance for both instantaneous and 3-hourly rain

We first investigate the effect of convective aggregation on the frequency of precipitation and on its distribution. The occurrence frequency of both instantaneous and 3-hourly precipitation over the domain is shown in Fig. 1 for both AGG and CTRL simulations. Instantaneous precipitation is not properly "instantaneous" but refers to an accumulation over a very short time corresponding to the model time step (less than 10 s). Any event is counted as rainy when instantaneous (resp. 3-hourly) precipitation is strictly positive. Non-zero thresholds were also tested without giving any significant changes. The CTRL simulation displays frequencies of precipitation rather uniform within the domain, for both instantaneous (Fig. 1a) and 3-hourly precipitation (Fig. 1b). As a reference, a perfect white noise sampled over a sufficiently long period of time would be perfectly uniform spatially. Frequencies of precipitation range from 0%-10% for instantaneous precipitation and reach 20%-30% when precipitation is accumulated over 3 hours. This increase of precipitation frequency with accumulation time is expected and reflects the increased likelihood of observing precipitation with longer timescales.

The differences are striking when one compares with the frequencies of the AGG simulation (Fig. 1c,d). While precipitation occurrence displays no particular pattern in the CTRL simulation, rainy events are concentrated on a circular core in the AGG simulation, where frequency of instantaneous precipitation can reach 60%. Conversely, the regions that are far from this circular core, exhibit very low occurrences of precipitation, often even zero (Fig. S1). Because no large-scale horizontal wind spontaneously develops in the domain, the moist region is almost steady in space, but its size slightly oscillates in time. In both simulations, the spatial distribution of deep convective clouds (and of intense cloud condensation rates) exhibits qualitatively similar patterns than those of precipitation shown in Fig. 1 (Figs. S2 and S3).

When one performs a spatial average of the precipitation frequencies over the whole

**Figure 2.** Probability density functions of instantaneous (a) and 3-hourly (b) precipitation intensity for aggregated (dashed) and unorganized (solid) simulations (Note that only rainy points are included here, which explains the different y-ranges, as 3-hourly precipitation is more often non zero than instantaneous precipitation).

Figure 3. Relative differences of instantaneous precipitation between the aggregated and the unorganized simulation as a function of instantaneous precipitation percentiles for rainy events. Dashed lines and grey shadings correspond to the signal  $\pm$  one standard deviation.

domain, it is found that there are fewer rainy events in the AGG simulation than in the CTRL simulation. Because of the RCE constraint, and similar domain-averaged radiative cooling rates between the two simulations, mean precipitation only slightly increases from the non-aggregated to the aggregated simulation (from 3.4 to 3.9 mm/day). Therefore, one could expect a general increase of rainfall rates (when strictly positive) in the AGG simulation compared to the CTRL. In other words, we find that precipitation is less frequent (from a frequency of 7.3% to 8.8% between both simulations) but heavier in the AGG simulation, for a similar domain-mean precipitation.

Differences in the number of rainy events between both simulations are related to the size of the convection core relative to the dry area in the AGG simulation. It is probable that model settings have an influence on the size of the aggregate and thus on rainfall frequencies. For instance, it has been shown that the ability of convection to selfaggregate in a cloud-resolving simulation is dependent on the domain size (Muller & Held, 2012; Jeevanjee & Romps, 2013). Nevertheless, we expect that qualitatively, the decreased frequency and increased intensity of precipitation with aggregation is robust.

We now turn to the full distribution of precipitation in those two simulations. Figure 2 displays the probability density functions (PDF) of instantaneous and 3-hourly precipitation for both CTRL and AGG simulations. Since our interest lies in precipitation intensity in regions where it rains and its response to aggregation, rather than differences in the area affected by precipitation, we consider the PDF of grid points with strictly positive instantaneous or 3-hourly precipitation. It corresponds to 7.3 % (15.5 %) of the total number of grid points in the AGG simulation against 8.8 % (29.8 %) in the CTRLsimulation for instantaneous (3-hourly) precipitation. Figure 2 shows an increase in the intensity of strong precipitation events, i.e. of the high precipitation percentiles, in the AGG simulation. The increase is more pronounced (in relative values) for 3-hourly precipitation (Fig. 2b) than for instantaneous precipitation (Fig. 2a). This is consistent with Fig. 1 which makes clear that rain accumulates locally over time in AGG, and with BS19 who show that time accumulated (daily in their analysis) precipitation is sensitive to the degree of aggregation. Such hypothesis could be verified using averaged rain intensities on hourly or sub-hourly outputs and computing the duration of rain events at these high percentiles, unfortunately that is not doable with the present data sampled every three hours.

Unlike the recent study of BS19, we find that instantaneous rainfall rates also increase significantly with aggregation, at a rate of about 30 % ± 8 % (here ± means one standard deviation for each percentile bin, computed following Fildier et al. (2018), equation (6) or appendix C, for estimating the  $\sigma$ -error on fractional changes between any pairs of simulations) as shown in Fig. 3. This figure shows relative differences of instantaneous precipitation as a function of precipitation percentiles. At a fixed percentile rank (e.g. the 99.9th percentile, the threshold in rain intensity for the most intense 1/1000 rainy points), this value increases by 30%, meaning that the most intense 1/1000 rainy points become 30% more intense. As previously, only instants with strictly positive instantaneous precipitation were selected for analysis. Albeit slightly lower, we also find significant increases of instantaneous precipitation extremes in the AGG simulation when including non-rainy instants (not shown).

The rest of the analysis will thus be devoted to understanding this significant and perhaps surprising increase of instantaneous extreme precipitation in the AGG simulation.

# 3.2 Decomposition into thermodynamic, dynamic, and microphysic contribution

The analysis is based on an adaptation of the O'Gorman and Schneider (2009a) scaling of precipitation extremes, which was used in many studies (Muller et al., 2011; Muller, 2013; Loriaux et al., 2017; Muller & Takayabu, 2020):

$$P = \epsilon \int -\rho w \frac{dq_{sat}}{dz} dz = \underbrace{\epsilon C}_{\text{scaling}}, \qquad (1)$$

where  $\epsilon$  is precipitation efficiency,  $\rho$  is air density, w is vertical velocity,  $q_{sat}$  is the saturated water vapor mixing ratio, the derivative is following a moist-adiabatic parcel ascent, and the integral is performed over positive vertical velocities, thus is a proxy of (gross) condensation rate (which will be replaced by C from now). In the tropics, the atmosphere is close to a moist adiabat and the parcel moves along the vertical to first order, therefore in practice, we used the local derivative

$$\frac{\partial q}{\partial z}$$

as an approximation for the derivative following a parcel ascent. This scaling can be interpreted from a water budget perspective, where a fraction  $\epsilon$  of the total condensation rate C reaches the ground as surface precipitation P. The remaining fraction of condensates either accumulates as clouds in the column, is advected away, or evaporates either as cloud or precipitation before reaching the ground. This scaling allows to distinguish the contributions to the fractional change in rain intensities, at any percentile, between the CTRL and AGG simulations:

$$\frac{\Delta P}{P} = \underbrace{\frac{\Delta \epsilon}{\epsilon}}_{\text{microphysic}} + \frac{\Delta C}{C}, \tag{2}$$

where  $\Delta$  denotes difference between AGG and CTRL (AGG minus CTRL) and denominators are calculated using the average of the CTRL and AGG values. We now focus on the second term and will investigate microphysical contributions (changes in precipitation efficiency) in section 5. We first separate the vertical mass flux contribution from the humidity contribution by neglecting second order terms:

$$\frac{\Delta C}{C} \approx \underbrace{\frac{\int \left(\frac{-dq_{sat}}{dz}\right)_{CTRL} \Delta \rho w \, dz}_{\text{dynamic}}}_{\text{dynamic}} + \underbrace{\frac{\int (\rho w)_{CTRL} \Delta \frac{-dq_{sat}}{dz} \, dz}_{\text{thermodynamic}}}_{\text{thermodynamic}} \tag{3}$$

The first term of the right-hand side is the change of condensation rate due to dynamic changes and the second term is the change of condensation rate due to thermodynamic changes.

This equation associates extreme precipitation values with a vertical profile of vertical velocities and saturation mixing ratio. Both of these profiles can be significantly modified with moist convection, and particularly through the formation of precipitation. Indeed, precipitation evaporation has a cooling effect which can both create downdrafts and reduce saturated mixing ratio. It is therefore not obvious that the vertical profile of vertical velocities and the one of the saturated mixing ratios when there is an extreme of surface precipitation, is representative of the earlier processes that led to this extreme.

Figure 4 displays composites of cloud condensate mixing ratio around points of extreme condensation rate, extreme cloud condensate mixing ratio (both vertically integrated), and extreme surface precipitation for both simulations. If one supposes that these three extremes correspond to a particular time of a same extreme precipitation event,

**Figure 4.** Composites of cloud condensate mixing ratio sections for extremes condensation rates (C; a,d), cloud condensate mixing ratio (b,e), and surface precipitation (Pr; c,f) for the aggregated (a,b,c) and unorganized (d,e,f) simulations. Extremes are defined by events above the 99<sup>th</sup> percentile. Vertical arrows shows vertical velocity. The horizontal axis represents the distance from the grid point that experienced the extreme, and the vertical axis is in pressure coordinates.

this figure illustrates the changes of clouds properties in between the time of these extremes. In fact, those composites are indeed consistent with the classical cloud lifecycle. First, concomitant with extreme condensation rate, upward motion is initiated in the low troposphere (Fig. 4a,d). Second, concomitant with extreme cloud amount, vertical updrafts reach the equilibrium level and cloud droplets grow in size and start to precipitate (Fig. 4b,e). Third, concomitant with extreme surface precipitation, latent cooling from rain evaporation generates intense downdrafts indicated by descending arrows and a higher cloud base above the extreme of surface precipitation (Fig. 4c,f). Extreme condensation rates were found in only 16% (CTRL) and 27% (AGG) of the clouds at this last stage (Fig. S4). To first order, the cloud lifecycle is well reproduced and visually similar in both simulations. Slight differences can be seen: in the aggregated case, condensate mixing ratios span a larger vertical range in the earlier stage of the lifecycle and a smaller vertical range later on; and vertical velocities are smaller in earlier stages and larger later, suggesting again that extreme rain events may last longer with aggregation. We note that this seems to appear when comparing composites, but individual events can strongly vary around the mean (Figs S5 and S6).

When evaluating the different contributions of the scaling in equations 2 and 3, the time chosen to evaluate variables that correspond to extreme surface precipitation is therefore important to have the best estimation for the scaling of precipitation extremes. The vertical velocity of extreme condensation prior to precipitation, is for instance the most relevant to extreme surface precipitation, as it is the effective upward motion that created the extreme of precipitation and which happened earlier in the cloud lifecycle. In the following, condensation rate, thermodynamic and dynamic contributions will be calculated at the extreme of condensation preceding extreme precipitation (section 4). Microphysic contributions will be calculated introducing extreme conversion rates (section 5) and diagnosed using variables calculated at extreme precipitation (e.g., for rain evaporation; section 5.1) and at extreme cloud amount (e.g., for autoconversion and accretion in evaporation; section 5.2).

## 4 Thermodynamic and dynamic contributions

This section investigates the contribution of the changes in condensation rates  $(\Delta C/C \text{ in eq. 2})$  to the changes in precipitation extremes seen in the previous section  $(\Delta P/P \text{ in eq. 2})$  between self-aggregated and unorganized convection. Figure 5 displays relative differences of condensation rate between the AGG and the CTRL simulations as well as its thermodynamic and dynamic contributions as a function of condensation rate percentiles. Since we selected only the rainy instants to assess the differences of extreme precipitation, we now select the cloudy instants (defined as instants for which the vertically integrated cloud mixing ratio is strictly positive) for assessing the contributions from both the thermodynamic and dynamic. These curves are calculated using mean of variables over 100 samples centered on one condensation rate percentile.

For the calculation of condensation rates (see equation 1), instead of computing the integral of the mean of the products (Fig. 5 in black), one could have computed the integral of the product of the means as represented in magenta in Fig. 5. While the forFigure 5. Relative differences of condensation rate ( $\Delta C$ , full black line, eq. 1) between the AGG and the CTRL simulations as well as their dynamic ( $\Delta \rho w$ , full blue line) and thermodynamic ( $\Delta \partial q/\partial z$ , full red line) contributions as a function of condensation rate percentiles (both computed following eq. 3). Dashed lines correspond to relative differences calculated between the 70<sup>th</sup> and 85<sup>th</sup> days whereas dot dashed lines correspond to relative differences calculated between the 85<sup>th</sup> and the 100<sup>th</sup> day. The magenta lines are the relative differences of condensation rate calculated as the integral of the product of mean (in each percentile bin) dynamical times mean thermodynamical terms ( $\Delta C_{approx}$ , unlike  $\Delta C$  obtained from the mean of the product).

Figure 6. Relative differences at the 99.9<sup>th</sup> percentile of: instantaneous precipitation from a direct calculation ( $\Delta$ Pr) and from the scaling used in eq. 3 ( $\Delta(\epsilon C)$ ), condensation rates ( $\Delta C$ ), precipitation efficiency ( $\Delta\epsilon$ ), conversion efficiency ( $\Delta\alpha$ ) with the contribution from accretion efficiency, and sedimentation efficiency ( $\Delta 1 - \beta$ ).

mer shows the actual condensation rate composited across all events, the latter is more adapted for comparison with the thermodynamic and dynamic contributions. This idea is similar to Fildier et al. (2018), showing that on the scale of a GCM grid box, the scaling in equation (1) can be computed equivalently before or after compositing on extreme rainfall events. Here, on convective scales, we see on Fig. 5 that both calculations give similar relative differences, indicating that non-linear terms do not act much, and thus validating our separation between dynamic and thermodynamic contributions.

Relative differences have been calculated over all the 30 days but also separately over the first 15 days and the last 15 days to evaluate the uncertainty (grey shadings in Fig. 5). It shows small changes between the first 15 and last 15 days for both contributions, giving us confidence that extremes have converged and that results are robust.

One can see a decrease of about - 20 % in extreme condensation rate in the AGG simulation compared to the CTRL simulation. This negative contribution is also shown in Fig. 6, which summarizes the main contributions to the changes in precipitation extremes between both simulations according to eq. 3. The decrease in extreme condensation rate is explained by a decreased dynamic contribution (- 25 %) in the AGG simulation, while the thermodynamic contribution remains similar or even slightly higher in the AGG simulation (+ 5 %).

# 4.1 Thermodynamic contribution driven by surface humidity changes

In order to explain positive and negative contributions of the thermodynamics and dynamics, changes in boundary layer water vapor mixing ratio (QBL) and Convective Available Potential Energy (CAPE) are displayed as a function of C and CAPE percentiles in Fig. 7a,b. Indeed, the former is related to the thermodynamic contribution, and the latter to the dynamic contribution (Muller & Takayabu, 2020). For consistency with the previous analysis, QBL was calculated only for cloudy instants.

One can observe an increase of about + 10 % of QBL in the AGG simulation for extreme C. This is qualitatively consistent, though slightly larger, than the thermodynamic contribution. A positive thermodynamic contribution is expected since self-aggregation is **Figure 7.** Relative differences in boundary layer water vapor mixing ratio (QBL, a) and Convective Available Potential Energy (CAPE, b) between the aggregated and the unorganized simulations as a function of condensation rate (C) and CAPE percentiles (resp.). Dashed lines correspond to relative differences calculated between the 70<sup>th</sup> and 85<sup>th</sup> days whereas dot dashed lines correspond to relative differences calculated between the 85<sup>th</sup> and the 100<sup>th</sup> day. Environmental and parcel vertical profiles of temperature for the aggregated and unorganized simulations (c) and differences of these profiles between both simulations (d), for events above the 99.9<sup>th</sup> percentile of CAPE.

associated with the confinement, and thus increase, of moisture in the convective region (Fig. 1). This increase in QBL is also consistent with a warming of a few degrees with aggregation if we assume the boundary layer saturated (which is a reasonable assumption in regions with high condensation rates).

As precised in Betts (1987); O'Gorman and Schneider (2009a), the moist-adiabatic derivative of saturation specific humidity does not increase (in absolute value) as quickly as does the saturated specific humidity with temperature. This may explain the higher rate of QBL changes compared to the thermodynamic contribution. This observation implies that in the process of cloud formation, part of the water vapor raised from the boundary layer remains not condensed at the top of the cloud, and particularly in the AGG simulation which is warmer and exhibits lower cloud tops than in the unorganized simulation (Fig. 7c,d).

Another potential reason explaining the discrepancy between the QBL relative differences and the value of the thermodynamic contribution is the limit of our method. Indeed, at the time of extreme C, QBL may not be representative of water vapor mixing ratio used for condensation which might result in unexplained differences. The scaling itself is also not perfect, local vertical variation of water vapor mixing ratio is only an approximation of the variation of water vapor mixing ratio that a parcel would experience when being raised: horizontal advection and losses by entrainment are not taken into account in this simplified scaling. Yet, we believe that the more humid boundary layer with aggregation is responsible for the increased thermodynamic contribution.

#### 4.2 Dynamic contribution driven by changes in local CAPE

Changes of CAPE between each simulation are shown as a function of CAPE percentiles since part of the CAPE is already consumed at extreme C, although this results in only small differences between each simulation at extreme C (not shown). Only strictly positive CAPE events were selected for analysis, in order to avoid the dry region of the AGG simulation (as done previously for precipitation, C, and QBL). At extreme CAPE, there is a decrease of about - 30 % of the CAPE in the AGG simulation, which would be consistent with a - 15 % decrease in extreme vertical velocities (proportional to the square root of CAPE). This value is thus smaller but qualitatively consistent with the dynamic contribution decrease of - 25 %. With the use of similar idealized simulations, Windmiller and Hohenegger (2019) showed that in such consecutively active regions, convection is preferentially triggered at the edge, where the CAPE is not maximal. This would imply that retaining the highest CAPE percentiles in the AGG simulation may overestimate the actual energy involved in the formation of extreme precipitation in this simulation, explaining the slight mismatch between the dynamic contribution and the one expected from CAPE differences at extreme CAPE percentiles.

The decrease in CAPE can be understood by considering temperature profiles of a

near-surface parcel rising adiabatically, and of its near-environment (Fig. 7c,d). The parcel temperature is somewhat similar with and without aggregation. The environmental temperature profile on the other hand is much warmer in the aggregated simulation. In other words, the atmosphere is closer to a moist adiabat with aggregation. This is expected, as with aggregation, convection and moisture are confined to a long-lasting moist region. So the entrained air at the edge of clouds is relatively moist, leading to a small net effect of turbulent entrainment at the edge of rising plumes (as the entrained air has properties close to the rising air; Singh and O'Gorman (2015)). Thus the atmospheric temperature is closer to an undilute moist adiabatic profile of the parcel, leading to smaller CAPE.

The extreme condensation rates are thus lower in the AGG simulation because of less favorable atmospheric instability, barely tempered by increased low-level moisture. These processes occur in the early stages of precipitation extreme formation and are followed by microphysical effects allowing the evolution of cloud drops to precipitating drops reaching the surface. The contribution of these effects to the enhanced precipitation extremes in the AGG simulation are discussed in the next section.

## 5 Microphysic contribution

This section investigates the contribution of the changes in precipitation efficiency  $(\Delta \epsilon / \epsilon \text{ in eq. 2})$  to the changes in precipitation extremes seen in section 3  $(\Delta P/P \text{ in eq. 2})$  between self-aggregated and unorganized convection. Figure 8 shows relative differences of precipitation efficiency between the AGG and the CTRL simulation as function of precipitation percentiles. Precipitation efficiencies were calculated dividing precipitation by a corresponding condensation rate. More precisely, bins of 100 samples were done for precipitation keeping only precipitation events and for condensation rates keeping only cloudy events. Both precipitation and condensation rate bins were classified in ascending order of precipitation or condensation rate (respectively). Since some clouds do not precipitate, the number of bins of condensation rate is higher than the one of precipitation. Therefore, the lowest condensate rate bins were removed and considered to not produce precipitation. It resulted in an equal number of bins for precipitation and condensation rate. Precipitation efficiencies were calculated dividing the mean of each precipitation bin by the mean of the corresponding condensation rate bin (i.e. percentile by percentile).

The precipitation efficiency is found to be the largest contribution to the change of precipitation extremes between both simulations, with an increase of + 50 % in the AGG simulation (Figs. 6 and 8). The values of precipitation efficiency at the 99.9<sup>th</sup> percentile reach about 0.48 in the AGG simulation against 0.28 in the CTRL simulation (Fig. S7; we note in passing that these values are within the range of existing estimates in the literature for similar simulations (Lutsko & Cronin, 2018; Fildier et al., 2020), though the precise values are somewhat sensitive to the microphysics and to the percentile of precipitation considered). In order to further investigate this change in precipitation efficiency, we follow Lutsko and Cronin (2018) and split the precipitation efficiency  $\epsilon$  into a term involving rain to surface precipitation efficiency  $(1-\beta)$ :

$$\epsilon = \frac{P}{C} = \frac{\int \rho q_p^{src} dz}{C} \times \frac{P}{\int \rho q_p^{src} dz} = \alpha \times (1 - \beta), \tag{4}$$

where  $q_p^{src}$  is the source (in kg kg<sup>-1</sup> s<sup>-1</sup>) of precipitating condensate amount  $q_p$  from the microphysics, and is directly diagnosed in the model outputs.

In other words,  $\alpha$  is the rate of conversion from cloud to precipitating condensate  $q_p^{src}$ , normalized by the rate of conversion from water vapor to cloud condensate C. Therefore,  $\alpha$  is called conversion efficiency (Lutsko & Cronin, 2018). The conversion efficiency  $\alpha$  increases by about +20 % with aggregation, for extreme precipitation events (Figs. 6 and 8). The extremes of  $q_p^{src}$  were evaluated at  $q_p^{src}$  percentiles to reflect the intermediate stage between the maximum in cloud condensation and the maximum in surface precipitation reached in a typical cloud lifecycle.

The other term,  $(1-\beta)$ , referred as sedimentation efficiency (Lutsko & Cronin, 2018), represents the fraction of source of precipitating condensate from microphysics which reaches the ground as surface precipitation. This fraction is typically less than unity be-

Figure 8. Relative differences of precipitation efficiency ( $\Delta \epsilon$ , full black line) between the AGG and the CTRL simulations as well as their contribution from the conversion of precipitation ( $\Delta \alpha$ , full dark red line) and from the evaporation of precipitation ( $\Delta \beta$ , full green line) as a function of surface precipitation percentiles. Dashed lines correspond to relative differences calculated between the 70<sup>th</sup> and 85<sup>th</sup> days whereas dot dashed lines correspond to relative differences calculated between the 85<sup>th</sup> and the 100<sup>th</sup> day.

**Figure 9.** Composites of (a,g,m) rain evaporation, (b,h,n) distance to saturation, (c,i,o) ventilation function  $f(v_t, D) = aD + bD\sqrt{Dv_t}$ , (d,j,p) rain mixing ratio, (e,l,q) graupel mixing ratio, (f,l,r) snow mixing ratio at times of extreme surface precipitation. The top panels show the CTRL simulation and the middle panels the AGG simulation in the vicinity of the extreme precipitation event (located at X = 0). The bottom panels show the profiles at the extreme precipitation location (i.e. at X=0).

cause of rain evaporation as precipitating condensates falls through subsaturated air. The increased sedimentation efficiency clearly dominates the precipitation efficiency increase, contributing approximately to + 30 % (Figs. 6 and 8).

In the next sections we investigate these changes of sedimentation  $(1-\beta)$  and conversion ( $\alpha$ ) efficiency between both simulations in more details.

#### 5.1 Changes in rain evaporation from changes in saturation deficit

The term  $1 - \beta$  is influenced by precipitation evaporation and precipitation transport (detrainment or entrainment). In the following, we assume that precipitation evaporation is the leading factor explaining the ratio  $1-\beta$ .

In SAM, the rate of precipitation evaporation

$$-\frac{\partial q_p}{\partial t} \propto (1 - RH) f(v_t, D)$$

is proportional to the distance to saturation 1 - RH and to a function  $f(v_t, D)$  of both the condensate terminal velocity  $(v_t)$  and the mean diameter of precipitation particles (D):  $f(v_t, D) = aD + bD\sqrt{Dv_t}$ . Where a and b can be considered as constants in our case. This latter term involves the so-called ventilation factor and accounts for the surface of precipitation particle available for evaporation. These variables, along with liquid, graupel and snow mixing ratios, are plotted in figure 9.

It shows that the evaporation of precipitation is less important in the aggregated simulation than in the unorganized simulation. The decrease in precipitation evaporation in the aggregated simulation is particularly significant in the lowest parts of the troposphere, where most of the precipitation evaporation occurs. Indeed, we found a decrease close to -50 % in the lowest 200 hPa of the atmosphere in the aggregated case. At these altitudes, the saturation deficit is about 68% less important in the aggregated simulation than in the unorganized simulation. Thus, the moister environment of the aggregated precipitation extremes largely explains the reduced precipitation evaporation near the surface. Finally, increased precipitation terminal velocity and diameter in the aggregated

simulation, tends to temper (through  $f(v_t, D)$ ) the differences of precipitation evaporation between both simulations by nearly 19% in the lowest 200 hPa of the troposphere (Fig. 9c,i,o).

Thus, consistent with BS19, we also find faster terminal velocities with aggregation, but in our case it is not due to a change in graupel. Instead, it is due to increased rain and decreased snow amounts (consistent with the warmer temperatures), with little change in graupel (Fig. 9d,e,f). BS19 further note that the faster terminal velocities, and associated shorter residence times, of rain compared to graupel or snow, imply reduced rain evaporation in their simulations. But for instantaneous precipitation rates, we believe that the instantaneous rain evaporation (Fig. 9a) is more relevant, which instead increases with terminal velocity (Fig. 9c,i,o) due to the aforementioned ventilation effect. This effect partially offsets the reduced rain evaporation from increased humidity (Fig. 9b,h,n).

#### 5.2 Larger conversion efficiency due to more efficient accretion

We noted that the conversion efficiency of extreme events  $\alpha$  increases by about 20% in the aggregated simulation. Since we also showed that extreme condensation rates decrease by 20% in the aggregated simulation, it implies that the extreme conversion rates remain similar in both simulations. This is indeed what we observe in average when comparing the vertical profiles of extreme conversion rates between both simulations (Fig. 10a,e,i), although the conversion rates are higher near the surface and smaller aloft for the aggregated simulation compared to the unorganized simulation.

In the 1-moment microphysics scheme that we use (Khairoutdinov & Randall, 2003), conversion from cloud to precipitating condensate is quantified as the sum of two microphysical processes:

• Accretion of precipitating type  $q_p$  due to collection of cloud condensate  $q_n$ :

$$\left(\frac{\partial q_p}{\partial t}\right)_{acc} \propto q_n q_p^a$$

where the exponent a depends slightly on the precipitation type (rain, graupel or snow in SAM) but is typically close to 1.

• Autoconversion and aggregation of cloud liquid and ice condensates (resp.)

$$\left(\frac{\partial q_p}{\partial t}\right)_{auto} \propto (q_n - q_{n,0})_+$$

where the coefficient of proportionality and threshold  $q_{n,0}$  are different for cloud liquid and cloud ice, and where subscript + indicates that negative values are replaced by zero.

Decomposing  $\alpha$  into contributions from both processes (through the decomposition of  $q_p^{src}$  into an accretion plus an autoconversion tendency in eq. 4), we see that the increase of  $\alpha$  in the aggregated simulation is almost entirely due to the first process, namely accretion, with a contribution of +18.7 % among the + 20% (Fig. 8b). We thus draw the vertical profiles of cloud condensates, precipitation, and their product as an estimation of the accretion rate vertical profiles for both simulations (Fig. 10b,c,d,f,g,h). The differences between the simulations change sign with altitude. In the highest altitudes (above 700 hPa), there is more cloud condensates and more precipitation in the unorganized simulation, which leads to higher accretion and conversion rates.

This increased condensate load aloft can be attributed to the increased condensation rates in the unorganized simulation. The result is an increase of auto-conversion rates at first which creates more precipitation particles, which combined with the increased condensate loads, provides a further increase of precipitation conversion through accretion in the unorganized simulation. The situation is different closer to the surface (in the lowest 200 hPa) where one can observe increased cloud condensates and precipitation particles, leading to increased accretion in the aggregated simulation. While we do not have evidence for this, the increased cloud condensates likely comes from the lower cloud bases in the aggregated simulation (a consequence of the increased surface relative humidity) and/or the Figure 10. Composites of (a,e,i) precipitation source from microphysics, (b,f,j) nonprecipitating condensates, (c,g,k) precipitating condensates, and (d,h,l)  $\rho QN \times \rho QP$  at times of extreme cloud amount. The top panels show the CTRL simulation and the middle panels the AGG simulation in the vicinity of the extreme event (located at X = 0). The bottom panels show the profiles at the extreme location (i.e. at X=0).

longer cloud lives in the aggregated simulation (as can be expected from the comparison of 3-hourly versus instantaneous precipitation differences in Fig 2). But the main increase of accretion rates near the surface arises from the increased precipitation explained by reduced evaporation in the aggregated simulation (Fig. 9a,g,m).

These differences between low and high altitudes tend to compensate each other and make averaged extreme accretion rates rather similar between both simulations. The effect is thus an increase of the accretion efficiency  $\alpha$  in the aggregated simulation.

## 6 Discussion and Conclusions

The spatial organization of deep-convective clouds is ubiquitous in the tropics and often leads to extreme weather. Importantly, one must question the extent to which models and idealized experiments can reproduce observed organized features and inform the rain changes that they induce. Two opposite types of organization are often studied in CRMs: regular cloud structures vs. convective clusters or moist patches, and metrics exist to test which type is more prevalent (Tompkins & Semie, 2017). Both organization types are sensitive to horizontal mixing in various ways (Piotrowski et al., 2009; Craig & Mack, 2013; Windmiller & Craig, 2019; Tompkins & Semie, 2017) and are controlled by different mechanisms occurring in the real atmosphere. By suppressing numerical dissipation, Piotrowski et al. (2009) observed that an original round feature of organized convection is not maintained, suggesting that some models cannot host convective organization without diffusion (numerical or through the turbulent scheme). Regular organization structures are controlled by cellular convection and cold pool interactions in the boundary layer (Feingold et al., 2010; Haerter, 2019). Instead, clustering, or aggregation, is driven by feedbacks that couple radiative cooling and convection, and wind effects on surface evaporation, leading to atmospheric instabilities in RCE (Emanuel et al., 2014; Bretherton et al., 2005). Here we focus on the latter, in a highly idealized manner. Although idealized self-aggregation of convection as modeled in square homogeneous RCE domains may be scarcely met in the current tropics, such states may be reached after a certain degree of warming (Wing & Emanuel, 2012) and some links with real world cloud clustering are already observable in current climate (Holloway et al., 2017). The present work does not address this question, but instead undertakes a detailed analysis of the mechanisms through which self-aggregation could affect precipitation extremes, once it occurs, if it occurs.

To do so, we analyze two idealized simulations using the SAM cloud-resolving model. In the first simulation, the atmosphere is organized into a moist convecting core and a dry convection-free region while in the second simulation convection is disorganized. We find that convective aggregation significantly increases precipitation extremes, by 70% for the  $99.9^{\rm th}$  percentile of 3-hourly precipitation consistent with earlier studies, but also surprisingly by a significant 30% increase for the  $99.9^{\rm th}$  percentile of instantaneous precipitation.

To investigate the processes responsible for the increase of instantaneous precipitation extremes with aggregation, we use an adapted scaling of precipitation extremes which accounts for the lifecycle of deep convective clouds. Previous studies using the scaling of precipitation extremes associated surface precipitation extremes with condensation rates and precipitation efficiencies co-located in time and space (O'Gorman & Schneider, 2009b; Muller et al., 2011; Muller, 2013; Loriaux et al., 2017). We argue that such spatiotemporal association may not be the most accurate method for investigating the causes of extreme precipitation, the latter significantly modifying the vertical profiles of thermodynamical and dynamical variables at their onset. Instead, one may evaluate the scaling variables a few minutes or hours before the extreme of precipitation (Fildier et al., 2020, e.g.). Although this alternative method would likely produce a more accurate relationship between precipitation, precipitation efficiency, and condensation rates, a bulk application is difficult since the location and time of the perfect scaling variables of one extreme precipitation event likely differ between these events. Here, we adopt another approach allowing to address this issue in a bulk manner and consisting in associating extreme precipitation percentiles with extreme percentiles of condensation and conversion rates, which mimicks different stages in the lifecycle of individual deep convective clouds. We use a decomposition of precipitation efficiency in a conversion and an evaporation term (as done in Lutsko and Cronin (2018)) and further investigate the role of the main microphysical variables. It is worth noting that such analyses could be reproduced in many other simulation or observational contexts.

Within this framework, we find that precipitation extremes are mainly increased through larger precipitation efficiency with convective aggregation. The contribution from the dynamics is largely negative with - 25 % while the thermodynamic contribution accounts for a comparatively small + 5 % contribution.

The increased thermodynamic contribution comes from enhanced moisture confinement in the boundary layer of the moist region for the case of aggregated convection, enabling convective updrafts to carry more water. The decreased dynamic contribution similarly derives from the moisture confinement in the moist convecting region, which reduces the effect of turbulent entrainment of environmental air at the edge of clouds. This results in an atmosphere closer to a moist adiabat, thereby reducing the strength of convective updrafts through a decrease of their buoyancy. Among these two opposite effects, the dynamical effect is the most important, and causes a 20% reduction of cloud condensation rates in convectively aggregated extreme events.

The microphysic contribution is the leading contribution to precipitation extremes enhancement, with a 50% increase of precipitation efficiency when convection is aggregated. Reduced rain evaporation contributes about 30% because of the moister environment of falling rain with aggregation. This is particularly true in the lowest 200 hPa where we find a 50% reduction of evaporation rates with aggregation. Increased accretion efficiency contributes another 20% increase in aggregated extreme precipitation.

The present results should be interpreted with care. It must be acknowledged that convective aggregation is sensitive to the modeling setup and forcing conditions chosen; this analysis is only one of a few starting points to study the response of extreme rain to changes in the organization of convection. Of particular importance is the sensitivity to the choice of modeling setup and subgrid-scale parameterizations, increasingly well-known and documented in the community. For instance, the degree of aggregation can be affected by the domain size and shape (rectangle vs. square, or long-channel domains; Cronin and Wing (2017)), thus affecting extreme precipitation intensities (Abbott et al., 2020). Coarse resolutions can also reduce the reevaporation of rain in downdrafts: that can inhibit cold pools and thus trigger aggregation (Jeevanjee & Romps, 2013). The representation of low clouds and the degree of convective mixing is also affected by the coarse resolutions and the turbulence scheme (Holloway et al., 2017; Tompkins & Semie, 2017) which affects the strength of aggregation, either through a moisture-memory feedback (Tompkins & Semie, 2017) or through a reduction in the radiative feedback (Muller & Held, 2012). This nonexhaustive list of modeling knobs that can trigger or inhibit aggregation is pursued by a whole field of research to further study the conditions in which self-aggregation can affect the real atmosphere, now and in the future.

Our results are qualitatively consistent with recent results from BS19, in the sense that the microphysics has a leading role in explaining extreme instantaneous precipitation in self-aggregated simulation. However, their idealized WRF simulations only exhibit a small change of instantaneous precipitation with aggregation. This may reflect the fact that microphysics is not well constrained in observations and models. Therefore, more work is desirable to investigate how precipitation efficiency affects precipitation extremes with convective aggregation, and how sensitive these results may be to the microphysics formulation in models. Three microphysics schemes exist in the System for Atmospheric Modeling used here: a single-moment scheme developed in the original version of the model and used for our simulations (Khairoutdinov & Randall, 2003), and two doublemoment microphysics schemes from Morrison et al. (2005) and Thompson et al. (2008). Knowing which microphysics scheme is most realistic should also be investigated, although the single-moment scheme used here has also been used in Global Cloud-Resolving Model experiments, resulting in model outputs that are visually indistinguishable from satellite observations (Stevens et al., 2019). The use of a spectral bin microphysics scheme would certainly provide a further step towards realism. Despite their high computational cost, such schemes have proven their better accuracy compared to bulk microphysics schemes (Khain et al., 2015). In an interest of making simulations closer to the real atmosphere, many other sensitivity tests must also be performed, by testing other parameterizations and model configurations as mentioned above.

The present work uses the surface-based CAPE to understand the decreased strength of convective updrafts when convection is aggregated. While we were mostly interested in a diagnosis rather than a precise estimation, it is worth noting that this metric may not be a good representation of every convective potentials. More elaborated metrics may be tested for that purpose such as the Most-Unstable CAPE, the mixed-layer CAPE, the generalized CAPE (Steinacker, 2017) or the Potential Energy Convertibility (Yano et al., 2005). The improvement of the metrics allowing to determine the potential intensity of convection is desirable for a better understanding of the changes in precipitation extremes with convective organization.

In a warming climate, one may expect an increase of precipitation extremes according to the Clausius-Clapeyron law which relates each degree of warming by an increase of about +7% of precipitation extremes. Our results suggest that this increase could be even more dramatic depending on the change of convective organization with temperature. Recent studies investigated the change of convective aggregation with SST under similar idealized settings. It is generally admitted that high SSTs favor convective self-aggregation (Pendergrass et al., 2016). Shamekh et al. (2020) showed that SST gradients are important as well for triggering convective aggregation. Overall the response of convective aggregation to warming remains uncertain and still an area of active research (Wing, 2019).

#### Acknowledgments

C.M. and B.F. gratefully acknowledge funding from the European Research Council (ERC) under the European Union's Horizon 2020 research and innovation programme (Project CLUSTER, grant agreement No 805041) and from the French national program Les Enveloppes Fluides et l'Environnement (LEFE) of Institut National des Sciences de l'Univers (INSU). The authors also thank Grand Équipement National De Calcul Intensif (GENCI), France, for providing access and support to their computing platforms Très Grand Centre de Calcul (TGCC). The simulation dataset is freely available at https://figshare.com/s/5ff1811271b1f3643004 (private link that will be made public when the article will be published). The authors would also like to thank the anonymous reviewers for their useful comments.

### References

- Abbott, T. H., Cronin, T. W., & Beucler, T. (2020). Convective Dynamics and the Response of Precipitation Extremes to Warming in Radiative–Convective Equilibrium. Journal of the Atmospheric Sciences, 77(5), 1637–1660. doi: 10.1175/jas-d-19-0197.1
- Attema, J. J., Loriaux, J. M., & Lenderink, G. (2014). Extreme precipitation response to climate perturbations in an atmospheric mesoscale model. *Environmental Research Letters*, 9(1), 014003. doi: 10.1088/1748-9326/9/1/014003

- Bao, J., & Sherwood, S. C. (2019). The role of convective self-aggregation in extreme instantaneous versus daily precipitation. Journal of Advances in Modeling Earth Systems, 11(1), 19–33. doi: 10.1029/2018MS001503
- Betts, A. K. (1987). Thermodynamic constraint on the cloud liquid water feedback in climate models. Journal of Geophysical Research: Atmospheres, 92(D7), 8483-8485. doi: 10.1029/JD092iD07p08483
- Bretherton, C. S., Blossey, P. N., Khairoutdinov, M., Bretherton, C. S., Blossey,
  P. N., & Khairoutdinov, M. (2005, dec). An Energy-Balance Analysis of Deep Convective Self-Aggregation above Uniform SST. Journal of the Atmospheric Sciences, 62(12), 4273–4292. doi: 10.1175/JAS3614.1
- Collins, W. D., Rasch, P. J., Boville, B. A., Hack, J. J., McCaa, J. R., Williamson,
  D. L., ... Zhang, M. (2006). The formulation and atmospheric simulation of the community atmosphere model version 3 (cam3). Journal of Climate, 19(11), 2144–2161. doi: 10.1175/JCLI3760.1
- Craig, G. C., & Mack, J. M. (2013). A coarsening model for self-organization of tropical convection. JGR, 118(16), 8761–8769.
- Cronin, T. W., & Wing, A. A. (2017, dec). Clouds, Circulation, and Climate Sensitivity in a Radiative-Convective Equilibrium Channel Model. Journal of Advances in Modeling Earth Systems, 9(8), 2883–2905. doi: 10.1002/2017MS001111
- Emanuel, K., Wing, A. A., & Vincent, E. M. (2014). Radiative-convective instability. Journal of Advances in Modeling Earth Systems, 75–90. doi: 10.1002/ 2013MS000270.Received
- Feingold, G., Koren, I., Wang, H., Xue, H., & Brewer, W. A. (2010). Precipitationgenerated oscillations in open cellular cloud fields. *Nature*, 466(7308), 849–852. doi: 10.1038/nature09314
- Fildier, B., Collins, W. D., & Muller, C. (2020). Distortions of the rain distribution with warming, with and without self-aggregation. Journal of Advances in Modeling Earth Systems, 12(e2020MS002256). doi: 10.1029/2020MS002256
- Fildier, B., Parishani, H., & Collins, W. D. (2018). Prognostic power of extreme rainfall scaling formulas across space and time scales. Journal of Advances in Modeling Earth Systems, 10(12), 3252-3267. doi: 10.1029/2018MS001462
- Haerter, J. O. (2019, apr). Convective Self-Aggregation As a Cold Pool-Driven

Critical Phenomenon. *Geophysical Research Letters*, 46(7), 4017–4028. doi: 10 .1029/2018GL081817

- Hamada, A., Takayabu, Y. N., Liu, C., & Zipser, E. J. (2015). Weak linkage between the heaviest rainfall and tallest storms. *Nature Communications*, 6, 6213. doi: 10.1038/ncomms7213
- Held, I. M., & Soden, B. J. (2006). Robust responses of the hydrological cycle to global warming. Journal of Climate, 19(21), 5686-5699. doi: 10.1175/JCLI3990.1
- Holloway, C. E., Wing, A. A., Bony, S., Muller, C., Masunaga, H., L'Écuyer, T. S.,
  ... Zuidema, P. (2017). Observing convective aggregation. Surveys in Geophysics, 38, 1199–1236. doi: 10.1007/s10712-017-9419-1
- Jeevanjee, N., & Romps, D. M. (2013). Convective self-aggregation, cold pools, and domain size. Geophysical Research Letters, 40(5), 994-998. doi: 10.1002/grl .50204
- Jiang, X., Adames, A. F., Kim, D., Maloney, E. D., Lin, H., Kim, H., ... Klingaman, N. P. (2020). Fifty years of research on the madden-julian oscillation: Recent progress, challenges, and perspectives. *Journal of Geophysical Research: Atmospheres*, 125(17), e2019JD030911. (e2019JD030911 2019JD030911) doi: 10.1029/2019JD030911
- Khain, A. P., Beheng, K. D., Heymsfield, A., Korolev, A., Krichak, S. O., Levin, Z.,
  ... Yano, J.-I. (2015). Representation of microphysical processes in cloud-resolving models: Spectral (bin) microphysics versus bulk parameterization. *Reviews of Geophysics*, 53(2), 247-322. doi: 10.1002/2014RG000468
- Khairoutdinov, M. F., & Randall, D. A. (2003). Cloud resolving modeling of the arm summer 1997 iop: Model formulation, results, uncertainties, and sensitivities. Journal of the Atmospheric Sciences, 60(4), 607-625. doi: 10.1175/1520-0469(2003)060{\textless}0607:CRMOTA{\textgreater}2.0.CO;2
- Loriaux, J. M., Lenderink, G., De Roode, S. R., & Siebesma, A. P. (2013). Understanding convective extreme precipitation scaling using observations and an entraining plume model. *Journal of the Atmospheric Sciences*, 70(11), 3641-3655. doi: 10.1175/JAS-D-12-0317.1
- Loriaux, J. M., Lenderink, G., & Siebesma, A. P. (2017). Large-scale controls on extreme precipitation. Journal of Climate, 30(3), 955–968. doi: 10.1175/JCLI

-18-

-D-16-0381.1

- Lutsko, N. J., & Cronin, T. W. (2018). Increase in precipitation efficiency with surface warming in radiative-convective equilibrium. Journal of Advances in Modeling Earth Systems, 10(11), 2992-3010. doi: https://doi.org/10.1029/ 2018MS001482
- Morrison, H., Curry, J. A., & Khvorostyanov, V. I. (2005, jun). A New Double-Moment Microphysics Parameterization for Application in Cloud and Climate Models. Part I: Description. Journal of the Atmospheric Sciences, 62(6), 1665–1677. doi: 10.1175/JAS3446.1
- Muller, C. (2013). Impact of convective organization on the response of tropical precipitation extremes to warming. *Journal of Climate*, 26(14), 5028-5043. doi: 10.1175/JCLI-D-12-00655.1
- Muller, C., & Bony, S. (2015). What favors convective aggregation and why? Geophysical Research Letters, 42(13), 5626-5634. doi: https://doi.org/10.1002/ 2015GL064260
- Muller, C., & Held, I. M. (2012). Detailed investigation of the self-aggregation of convection in cloud-resolving simulations. Journal of the Atmospheric Sciences, 69(8), 2551–2565. doi: 10.1175/JAS-D-11-0257.1
- Muller, C., O'Gorman, P. A., & Back, L. E. (2011). Intensification of precipitation extremes with warming in a cloud-resolving model. Journal of Climate, 24(11), 2784-2800. doi: 10.1175/2011JCLI3876.1
- Muller, C., & Takayabu, Y. (2020). Response of precipitation extremes to warming: what have we learned from theory and idealized cloud-resolving simulations, and what remains to be learned? *Environmental Research Letters*, 15(3), 035001. doi: 10.1088/1748-9326/ab7130
- O'Gorman, P. A., & Schneider, T. (2009a). The physical basis for increases in precipitation extremes in simulations of 21st-century climate change. *Proceedings* of the National Academy of Sciences, 106(35), 14773–14777. doi: 10.1073/pnas .0907610106
- O'Gorman, P. A., & Schneider, T. (2009b). Scaling of precipitation extremes over a wide range of climates simulated with an idealized gcm. *Journal of Climate*, 22(21), 5676–5685. doi: /10.1175/2009JCLI2701.1

Pendergrass, A. G., Reed, K. A., & Medeiros, B. (2016). The link between ex-

-19-

treme precipitation and convective organization in a warming climate: Global radiative-convective equilibrium simulations. Geophysical Research Letters, 43(21), 11,445-11,452. doi: https://doi.org/10.1002/2016GL071285

- Piotrowski, Z. P., Smolarkiewicz, P. K., Malinowski, S. P., & Wyszogrodzki, A. A. (2009). On numerical realizability of thermal convection. *Journal of Computational Physics*, 228(17), 6268–6290. doi: 10.1016/j.jcp.2009.05.023
- Romps, D. M. (2011). Response of tropical precipitation to global warming. Journal of the Atmospheric Sciences, 68(1), 123-138. doi: 10.1175/2010JAS3542.1
- Rossow, W. B., Mekonnen, A., Pearl, C., & Goncalves, W. (2013). Tropical precipitation extremes. *Journal of Climate*, 26(4), 1457–1466.
- Shamekh, S., Muller, C., Duvel, J.-P., & D'Andrea, F. (2020). How do ocean warm anomalies favor the aggregation of deep convective clouds? Journal of the Atmospheric Sciences, 77(11), 3733 - 3745. doi: 10.1175/JAS-D-18-0369.1
- Singh, M. S., & O'Gorman, P. A. (2015). Increases in moist-convective updraught velocities with warming in radiative-convective equilibrium. Quarterly Journal of the Royal Meteorological Society, 141(692), 2828–2838. doi: 10.1002/qj.2567
- Singleton, A., & Toumi, R. (2013). Super-clausius-clapeyron scaling of rainfall in a model squall line. Quarterly Journal of the Royal Meteorological Society, 139(671), 334-339. doi: 10.1002/qj.1919
- Steinacker, R. (2017, mar). A New Look to the Generalized CAPE. Journal of the Atmospheric Sciences, 74(3), 767–781. doi: 10.1175/JAS-D-15-0210.1
- Stevens, B., Satoh, M., Auger, L., Biercamp, J., Bretherton, C. S., Chen, X., ... others (2019). Dyamond: the dynamics of the atmospheric general circulation modeled on non-hydrostatic domains. *Progress in Earth and Planetary Science*, 6(1), 1–17.
- Tan, J., Jakob, C., Rossow, W. B., & Tselioudis, G. (2015). Increases in tropical rainfall driven by changes in frequency of organized deep convection. *Nature*, 519(7544), 451. doi: 10.1038/nature14339
- Thompson, G., Field, P. R., Rasmussen, R. M., & Hall, W. D. (2008, dec). Explicit Forecasts of Winter Precipitation Using an Improved Bulk Microphysics Scheme. Part II: Implementation of a New Snow Parameterization. Monthly Weather Review, 136(12), 5095–5115. doi: 10.1175/2008MWR2387.1

-20-

- Tompkins, A. M., & Semie, A. G. (2017, jun). Organization of tropical convection in low vertical wind shears: Role of updraft entrainment. Journal of Advances in Modeling Earth Systems, 9(2), 1046-1068. doi: 10.1002/2016MS000802
- Windmiller, J. M., & Craig, G. C. (2019). Universality in the spatial evolution of self-aggregation of tropical convection. Journal of the Atmospheric Sciences, 76(6), 1677-1696. doi: 10.1175/JAS-D-18-0129.1
- Windmiller, J. M., & Hohenegger, C. (2019).Convection on the edge. Journal of Advances in Modeling Earth Systems, 11(12), 3959-3972. doi: 10.1029/ 2019MS001820
- Wing, A. (2019).Self-aggregation of deep convection and its implications for climate. CCCR, 5(1), 1-11.
- Wing, A., & Emanuel, K. (2012, apr).Organization of Tropical Convection: Dependence of Self-Aggregation on SST in an Idealized Modeling Study. paper presented at 30th Conference on Hurricanes and Tropical Meteorology, Am. Meteorol. Soc., Ponte Vedra Beach, Fla.
- Wing, A., Emanuel, K., Holloway, C., & Muller, C. (2017).Convective selfaggregation in numerical simulations: A review. Surv. Geophys., 38, 1173-1197. doi: 10.1007/s10712-017-9408-4
- Yano, J.-I., Chaboureau, J.-P., & Guichard, F. (2005, apr). A generalization of CAPE into potential-energy convertibility. Quarterly Journal of the Royal Meteorological Society, 131(607), 861–875. doi: 10.1256/QJ.03.188